Author’s response to reviews

Title: Proteinuria as a presenting sign of combined methylmalonic acidemia and homocysteinemia: case report

Authors:
Ruyue CHEN (781135340@qq.com)
Xiao-zhong LI (xiaozhonglicn@yeah.net)
Qiang LIN (wslinqiang@aliyun.com)
Yun ZHU (ruyuechen@foxmail.com)
Yun-yan SHEN (syunyan@sohu.com)
Qin-ying XU (listenway_li@163.com)
Xue-ming ZHU (xueming_zhu@aliyun.com)
Lin-Qi CHEN (clq631203@aliyun.com)
Hai-Ying WU (hainewhy@126.com)
Xu-Qin CHEN (1136051418@qq.com)

Version: 7 Date: 09 Sep 2020

Author’s response to reviews:

Dear Editor,

We are grateful for the suggestions provided by you for our manuscript entitled “Proteinuria as a presenting sign of combined methylmalonic acidemia and homocysteinemia: case report”. The datasets have been deposited in NCBI and are public. You can find it according to these four numbers (SRR12586543, SRR12586542, SRR12586541, and SRR12586540). PRJNA661347 is BioProject number, not the accession number. These materials described in the manuscript, including all relevant raw data are available upon request. We hope that the revised version will be acceptable for publication. We look forward to hearing from you at your earliest convenience.

Yours faithfully
Corresponding author: Xiao-Zhong LI.